Author's response to reviews

Title: The changing patterns of dispensing branded and generic drugs for the treatment of gastroesophageal reflux disease between 2006 and 2011 in Japan: a retrospective cohort study

Authors:

Kyoko Murata (murata.kyoko.5m@kyoto-u.ac.jp)
Shiro Hinotsu (hinotsus@cc.okayama-u.ac.jp)
Shota Hamada (hamada.shota.4m@kyoto-u.ac.jp)
Yasumasa Ezoe (yasuzoe@kuhp.kyoto-u.ac.jp)
Manabu Muto (mmuto@kuhp.kyoto-u.ac.jp)
Koji Kawakami (kawakami.koji.4e@kyoto-u.ac.jp)

Version: 2
Date: 2 July 2014

Author's response to reviews: see over
Dear Editors,

*BMC Health Services Research*

MS: 1776710153134212

We revised our manuscript to include line numbers based on the editorial request, received on 1st July. We hope our revision would be sufficient to proceed with peer review.

We appreciate your consideration of our manuscript for publication in *BMC Health Services Research*.

Sincerely yours,

Koji Kawakami, MD, PhD
Professor and Chairman
Department of Pharmacoepidemiology, Graduate School of Medicine and Public Health, Kyoto University, Yoshidakonoecho, Sakyoku, Kyoto, Japan 606-8501
Phone: +81-75-753-4459, Fax: +81-75-753-4469
E-mail: kawakami.koji.4e@kyoto-u.ac.jp